

Dear Science Editor, Company Editor-in-Chief,

Greetings of the day,

I would like to thank you very much for your support, and I'm very pleased to get profit from your recommendations and pertinent comments to improve the quality of my manuscript.

Thank you again,

Warm regards

Dear Reviewer,

I would like to express my sincere gratitude to you for your kind attention and pertinent recommendations.

I'm happy to return to you my revised manuscript with the highlighted changes.

I really benefited from your comments. In fact, i already integrated genetic discussion and examination in my manuscript that you may find Highlighted. For that, I added some references that the numbers of the references have changed.

Thank you again,

Warm regards.